Author's response to reviews

Title: Physician attitude toward depression care interventions: implications for implementation of quality improvement initiatives

Authors:

Rachel EM Henke (rhenke@post.harvard.edu)
Ann F Chou (ann-chou@ouhsc.edu)
Johann C Chanin (chanin@ncqa.org)
Amanda B Zides (aberke23@aol.com)
Sarah H Scholle (scholle@ncqa.org)

Version: 4 Date: 13 September 2008

Author's response to reviews:

Hi,

I am attaching the revised manuscript. Let me know if there is anything else I need to do before publication. Thanks!

Rachel